Author's response to reviews

Title: Waist circumference: as an indicator of risk for type 2 diabetes and cardiovascular disease

Authors:

Reijo Siren (reijo.siten@fimnet.fi)
Johan G Eriksson (johan.eriksson@helsinki.fi)
Hannu Vanhanen (Hannu.vanhanen@kela.fi)

Version: 3 Date: 20 June 2012

Author's response to reviews: see over
Dear Editor

We would like to thank the reviewers and the editor for their constructive criticism and comments. Please find below our itemized response to the reviewer’s comments

Reviewer Diewertje Sluik

1. We have included that information.
2. We have included that information.
3. We have changed that part of the conclusion according to the suggestions by the reviewer.

Reviewer Cia-Hin Lau

No further comments

We hope that you will find the revised version suitable for publication.

Sincerely

Johan G Eriksson

MD, DMSc
Professor
University of Helsinki